Author's response to reviews

Title: Transversus Abdominis Plane Block or Local Anesthetic Infiltration? A Meta-Analysis of Randomized Controlled Trials

Authors:

Nanze Yu (yunanze@hotmail.com)
Xiao Long (pumclongxiao@126.com)
Jorge R Lujan-Hernandez (Jorge.lujan87@hotmail.com)
Julien Succar (julas19@gmail.com)
Xin Xin (xinxin.harvard@gmail.com)
Xiaojun Wang (xjwang100@hotmail.com)

Version: 2 Date: 27 March 2014

Author's response to reviews:

Line and page numbers are included in the manuscript.